Author's response to reviews

Title: Treatment of hyperprolactinemia: a systematic review and meta-analysis

Authors:

Amy T Wang (wang.amy@mayo.edu)
Rebecca J Mullan (mullan.rebecca@gmail.com)
Melanie A Lane (melanie7@gmail.com)
Ahmad Hazem (adzhazem@yahoo.com)
Chaithra Prasad (prasad.chaithra@mayo.edu)
Nicola W Gathaiya (gathaiya.nicola@mayo.edu)
M Merce Fernandez-Balsells (mercefernandez.girona.ics@gencat.cat)
Amy Bagatto (aimsjoy@gmail.com)
Fernando Coto-Yglesias (cotoyglesias@yahoo.com)
Jantey Carey (janteycarey@gmail.com)
Tarig A Elraiyah (elraiyah.tarig@mayo.edu)
Patricia J Erwin (erwin.patricia@mayo.edu)
Gunjan Y Gandhi (gandhi.gunjan@mayo.edu)
Victor M Montori (montori.victor@mayo.edu)
Mohammad H Murad (murad.mohammad@mayo.edu)

Version: 3 Date: 20 March 2012

Author's response to reviews: see over
March 20, 2012

Paul G. Shekelle, MD, MPH, PhD
David Moher, PhD
Lesley A Stewart, PhD
Editors-in-Chief
Systematic Reviews
c/o BioMed Central
236 Gray's Inn Road
London WC1X 8HB
United Kingdom

Dear Drs. Shekelle, Moher, and Stewart

On behalf of my co-authors, we would like to thank you and your editorial staff for your consideration of our manuscript entitled “Treatment of hyperprolactinemia: a systematic review and meta-analysis.”

This body of work provides evidence to guide practitioners in providing therapy for their patients with hyperprolactinemia. The search has been updated as of December 15, 2011

This manuscript has been written, revised, and approved by all listed co-authors. We believe all the information contained within to be accurate and true. We hereby confirm that neither the manuscript, nor any part of it, has been published or is being considered for publication elsewhere. Each of us acknowledges that he or she participated sufficiently in the work to take public responsibility for its content.

We have no conflicts of interest or other relationships which might lead to a conflict of interest.

We appreciate your consideration of this manuscript for publication in the Systematic Reviews Journal and look forward to responding to any additional questions that you or your editorial staff may have.

Sincerely yours,

Amy Tu Wang, MD
Mayo Clinic
200 First Street SW
Rochester MN 55905
wang.amy@mayo.edu
Phone: 507-284-9994
Fax: 507-284-4959